Author's response to reviews

Title: Protocol for "Seal or Varnish?" (SoV) trial: a randomised controlled trial to measure the relative cost and effectiveness of pit and fissure sealants and fluoride varnish in preventing dental decay.

Authors:

Ivor G Chestnutt (chestnuttig@cardiff.ac.uk)
Barbara L Chadwick (chadwickbl@cardiff.ac.uk)
Simon Hutchings (hutchingss1@cardiff.ac.uk)
Rebecca Playle (playlera@cardiff.ac.uk)
Timothy Plckles (pickleste@cardiff.ac.uk)
Catherine Lisles (lislesca1@cardiff.ac.uk)
Nigel KIrby (Kibryn@cardiff.ac.uk)
Maria Z Morgan (morganmz@cardiff.ac.uk)
Lindsay Hunter (hunterml@cardiff.ac.uk)
Ceri Hodell (ceri.hodell@wales.nhs.uk)
Beverely Withers (beverely.withers@wales.nhs.uk)
Simon Murphy (murphys7@cardiff.ac.uk)
Sarah Morgan-Trimmer (morgan-trimmersa@cardiff.ac.uk)
Deborah Fitzsimmons (D.Fitzsimmons@swansea.ac.uk)
Ceri Phillips (c.j.phillips@swansea.ac.uk)
Jacqueline Nuttall (nuttallj@cardiff.ac.uk)
Kerenza Hood (hoodk1@cardiff.ac.uk)

Version: 2  Date: 9 November 2012

Author's response to reviews:

Dear Dr Foote,

Thank you for your e-mail relating to our submission. In response to the points raised:

1 I can confirm that there are no other manuscripts arising from this study under consideration at other journals

2 I have added the Trial registration numbers to the end of the abstract and also on the submission system.

3 I have removed the supplementary files containing proof of research ethics approval and of grant funding by NIHR.

4 I have removed the Figures from within the main manuscript and left this as just the 3 separate Word files.

I trust that this is all in order and thank you for your swift review and handling of our protocol.
Kind regards
Ivor Chestnutt